Author's response to reviews

Title: Oesophageal Perforation: A diagnostic and therapeutic challenge in a resource limited setting. A report of three cases

Authors:

Deo D Balumuka (balumukad@gmail.com)
Philipo L Chalya (drphillipoleo@yahoo.com)
William Mahalu (William.mahalu@yahoo.com)

Version: 6 Date: 22 September 2011

Author's response to reviews:

dear, sir,

the address of the institution is written as p box...
i have changed it to p.o.box....
that will be my final change the rest is well.

thank you very much, sir.

dr.balumuka